Author's response to reviews

Title: Collaborating with consumer and community representatives in health and medical research in Australia: results from an evaluation

Authors:

Janet M Payne (janp@ichr.uwa.edu.au)
Heather A D’Antoine (heather.dantoine@menzies.edu.au)
Kathryn E France (k.france@ecu.edu.au)
Anne E McKenzie (annem@ichr.uwa.edu.au)
Nadine Henley (n.henley@ecu.edu.au)
Anne E Bartu (A.E.Bartu@curtin.edu.au)
Elizabeth J Elliott (ElizabE2@chw.edu.au)
Carol Bower (carolb@ichr.uwa.edu.au)

Version: 3 Date: 25 April 2011

Author's response to reviews: see over
25th April 2011

The Editors-in-Chief
Health Research Policy and Systems
BioMed Central Ltd
Floor 6, 236 Gray’s Inn Road
London, WC1X 8HB

Dear Editors-in-Chief

Collaborating with consumer and community representatives in health and medical research in Australia: results from an evaluation
MS ID: 874864144982415

I herewith submit the changes that you have requested to the format of the above manuscript.

Major revisions made:

Authors’ contributions: Initials have been used instead of full name.

References: functioning links have been provided to websites 42 and 48.

Minor revisions made:

Title page: has been arranged in the following order: Title; author list; affiliations; corresponding author details; email addresses.

Corresponding author: one corresponding author(s) has been indicated on the title page, and email address given.

Abstract: we now only mention http://www.ichr.uwa.edu.au/alcoholandpregnancy once in the abstract.

Abstract: word count has been removed from the abstract.

Highlighting/tracking: All highlighting and tracking has been removed from the manuscript.

Typography: The manuscript has been checked for any typographical errors and two final corrections made. (1) On page 8, background, last paragraph, line 2, “consumer and” has been deleted; and (2) on page 35, reference, the reference has been formatted according to the journal style.

Thank you.

Yours sincerely

Janet M Payne
Senior Research Officer